Author's response to reviews

Title: The role of diagnostic ureteroscopy in the era of computed tomography urography.

Authors:

Shay Golan (shaygo1@gmail.com)
Andrei Nadu (andrei.nadu@sheba.health.gov.il)
David Lifshitz (davidlif@clalit.org.il)

Version: 4 Date: 22 June 2015

Author's response to reviews: see over
June 22nd, 2015

To:
BioMed Central
236 Gray's Inn Road
London, United Kingdom
Telephone +44 (0) 20 3192 2013
E-mail: editorial@biomedcentral.com

Re: MS: 4421673711349165, “The role of diagnostic ureteroscopy in the era of computed tomography urography”

Dear Dr. Grande,

Thank you for your letter of June 21st regarding the above-referenced manuscript. Our point-by-point reply appears below.

1. The text was modified accordingly [Lines 109-111]:
   "Because of the retrospective design of this study, our institutional ethics committee waived the need for written informed consent from participants".

2. The acknowledgements section has been modified to clarify that no financial support was received for this study.

3. Formatting guidelines have been fulfilled.

4. STROBE reporting guidelines have been fulfilled.

We hope you find the revised version of our manuscript suitable for publication.
Sincerely,

Shay Golan, MD